Author's response to reviews

Title: Developing program theory for purveyor programs

Authors:

Christa Oosthuizen (me.coosthuizen@gmail.com)
Johann Louw (Johann.Louw@uct.ac.za)

Version: 8 Date: 12 February 2013

Author's response to reviews:

The authors changed the figures to conform to US English. Programme was changed to program in both figures. The copy editor did not make these changes, but we assumed that everything should be US English.